Author’s response to reviews

Title: Progesterone after previous preterm birth for prevention of neonatal respiratory distress syndrome (PROGRESS): a randomised controlled trial.

Authors:

Jodie M Dodd (jodie.dodd@adelaide.edu.au)
Caroline A Crowther (caroline.crowther@adelaide.edu.au)
Andrew J McPhee (mcpheea@wch.sa.gov.au)
Vicki Flenady (vicki.flenady@mater.org.au)
Jeffrey S Robinson (jeffrey.robinson@adelaide.edu.au)

Version: 2 Date: 30 January 2009

Author’s response to reviews:

Please find attached an updated version of the manuscript. I have attended to the changes as requested below. The word count of the abstract is now 350 words; the names of research & ethics committees approving the study to date have been included in the methods; the abstract contains the trial registration details; we have added a statement about competing interests.